The evolution of herd behavior: will herding disappear over time?

Kalugala Vidanalage Aruna Shantha

School of Management, Wuhan University of Technology, Wuhan, China and Department of Accountancy and Finance, Faculty of Management Studies, Rajarata University of Sri Lanka, Mihinthale, Sri Lanka

Abstract

Purpose – The purpose of this paper is to examine the evolutionary nature of herding phenomenon in the context of a frontier stock market, the Colombo Stock Exchange of Sri Lanka.

Design/methodology/approach – This study applies the cross-sectional absolute deviation methodology for daily frequencies of data of all the common stocks listed during the period from April 2000 to March 2018. The regression coefficients are estimated by using both the ordinary least square and the quantile regression procedures.

Findings – The findings reveal significant changes to the pattern of herding over different market periods, each with specific characteristics. Herding is strongly evident in up and down market days in the 2000-2009 period, during which the market was highly uncertain with the impact of the political instability of the country due to the Civil War on the stock trading. Even after this Civil War period, herd tendency is strongly manifested toward the up market direction as a result of the investors' optimism about the country's economy and political stability, which caused to a speculative bubble in the market. After that, it is turned into negative herding due to the panic selling occurred in view of the uncertainty of the inflated prices, which led to a market crash. Notably, herding appears to be consistently absent over the period after the crash, despite the presence of herd motives such as high market uncertainties triggered by political instability and economic crisis during that period.

Research limitations/implications – The findings suggest that herd behavior is an evolving phenomenon in financial markets. Consistent with the adaptive market hypothesis, the absence of herding evident after the market crash could be attributed to the investors' learning of the irrationality of herding/ negative herding for adapting to market conditions. As a result, herding and negative herding tendencies declined and disappeared at the aggregate market level.

Originality/value – This study contributes to the literature by providing novel evidence on the evolutionary nature of behavioral biases, particularly herding, as predicted by the adaptive market hypothesis. With the application of the quantile regression procedure, in addition to customary used ordinary least squares approach, it also provides robust evidence on this phenomenon.

Keywords Quantile regression, Adaptive market hypothesis, Colombo stock exchange, Frontier market, Herd evolution, Market bubble and crash, Investor learning, Negative herding

Paper type Research paper

1. Introduction

Behavioral economics, a branch of economics that blends insights of psychology and economics, attempts to explore how, when and why individuals behave irrationally when making decisions. The theories and concepts of behavioral economics have greatly contributed to the development of behavioral finance which strives to find answers for

JEL classification - C12, C31, G14, G41

The author would like to thank the editor, Wing-Keung Wong and the anonymous referees for helpful comments and suggestions.

Studies in Economics and Finance Vol. 36 No. 4, 2019 pp. 637-661 © Emerald Publishing Limited 1086-7376 DOI 10.1108/SEF-06-2018-0175

Received 12 June 2018 Revised 28 September 2018 24 November 2018 25 February 2019 6 June 2019 Accepted 20 June 2019

Evolution of herd behavior

the irrationality of the participants in financial markets. Over the past few decades, the behavioral finance researchers have dedicated appreciable attention for understanding the behavior of market participants and biases associated with their actions when dealing with financial assets. Their findings reveal that such biases occur due to limitations of the inner cognitive abilities and psychological states as well as effects of external stimuli, for example, market conditions, social structures and information asymmetry. These findings, while challenging the "rational expectation" notion of the efficient market hypothesis, support the prediction of behavioral finance that the market participants' behavior is bounded rational, which results to maladaptive decisions and eventually leads to poor performance. Based on the concept of bounded rationality and some principles of evolutionary biology and evolutionary psychology, Lo (2004, 2005, 2012) introduced a new paradigm called "adaptive market hypothesis" (AMH) to reconcile this debate between the efficient market hypothesis and the behavioral finance.

The AMH assumes that the rationality of market participants' behaviors, thereby, efficient functioning of a financial market is an evolving phenomenon, driven by market forces and the market participants' adaptability to the environment in which they behave. According to this theory, investment strategies (heuristics) become biased or irrational when they do not fit with the context to which they apply. Their impacts to market prices depend on the proportion of the population with biased heuristics verses the proportion of the competing population with more effective heuristics that are adaptive to market conditions. The evolutionary perspective of this theory implies that competition drives the market participants who are susceptible to biases, to learn from their biases for adapting to the market conditions. Accordingly, given enough time and competitive market forces, the market participants who cannot adapt will exit from the market due to losses. Thus, the market will eventually consist of those who adapt to its conditions with appropriate heuristics. Consequently, behavioral biases would decline in the market and, thereby, prices would approach to their rational level.

One of the behavioral biases extensively discussed in the behavioral finance literature is herding. It is one's propensity to abandon own information and belief and imitate the actions of others in making choices. This behavioral tendency has been vastly studied in both developed and emerging financial markets. However, it has limitedly been explored in respect of frontier stock markets, a category of markets which is predicted to be more vulnerable for that when compared to developed and emerging markets (Economou, 2016; Guney *et al.*, 2017). Further, based on the evolutionary perspective of the AMH, herding would follow an evolving path, in the sense that it would tend to decline and disappear through the competition and the adaptation of market participants' behaviors. However, this evolution phenomenon, to the best of my knowledge, has not been much supported with empirical evidence at the aggregate market level. Therefore, this study aims to explore the evolution of herding in the stock market of Sri Lanka, a frontier market[1], formally known as the Colombo Stock Exchange (CSE).

The CSE is an ideal ground for the examination of the evolution of herding due to the following reasons. First, as discussed in Section 3, it, as a pre-emerging market, is small and illiquid, as characterized by low capitalization, low trading volume and relatively few number of stocks actively traded, which could discourage investor participation for investing and trading activities. In addition, the market is regularly exposed to certain risks that may cause incidents such as information asymmetry, insider trading, low transparency, price manipulations and significant mispricing. These conditions would impair both the amount and the trustworthiness of the information generated, eventually rendering to a more ambiguous informational environment in the market. As a result, the investors would

SEF

36.4

tend to herd for resolving this information uncertainty (Guney *et al.*, 2017). Second, over the last two decades, the trading environment of the CSE has been highly uncertain owing to incidents of political instability and economic crises of the country. Given such stressful environment, it is also more probable that the investors disregard their own beliefs and are prone to imitate others' behaviors when making their trading decisions (Arjoon and Bhatnagar, 2017; Hwang and Salmon, 2004; Olsen, 2011). Third, the previous studies on the CSE find both the evidence of herding and the absence of it over different market periods (Sewwandi, 2016; Xiaofang and Shantha, 2017, 2018) and the investors' tendency to learn the irrationality of such behavior (Shantha, 2019), which can be considered as some clues of the evolution of herd behavior in the market.

Accordingly, a longer sample period, from April 2000 to March 2018 which cover different uncertain episodes of the CSE (Figure 1), was chosen in this study in an attempt to detect herd evolution in the market. The findings reveal that herding, which triggered during the period of political instability due to Civil War and the market bubble period, turns into negative herding during the subsequent market crash period. It, however, appears to be consistently absent during the period after the crash despite the presence of herd motives such as market uncertainties due to the political instability and economic crisis of the country during this period. Hence, these results validate the prior evidence obtained at the individual investor level that indicates the investors' tendency to get away from herding by learning its irrationality from the experiences of losses and associated psychological shocks (Shantha, 2019). Accordingly, the findings provide a strong support for the evolutionary nature of behavioral biases, as implied by the AMH.

The findings of this work contribute the literature in the following ways. First, this study addresses a relatively underexplored behavioral finance concern on the evolutionary nature of behavioral biases, as predicted by the AMH. Hence, while supporting this evolutionary perspective, the findings will enhance the understanding of the nature of behavioral biases, particularly herding phenomenon in financial markets. Second, the previous studies conducted at the aggregate market level suggest the regulatory reforms subsequent to a market crisis period as a turning point for herd evolution[2]. On the contrary, the findings of this study reveal the absence of herding during the period after the market crash even though the market was not undergone





herd behavior

Evolution of

Note: This figure plots the movements of the All share price index, a value-weighted index representing all the stocks listed on the CSE, over the period from April 2000 to March 2018t

SEF significant regulatory reforms over that period. Thus, with the support of prior evidence reported at the individual investor level (Shantha, 2019), the results imply the investors' 36.4 mistake-learning behavior as another possible way of herd evolution in financial markets. Third, unlike previous studies conducted on the CSE (Sewwandi, 2016; Xiaofang and Shantha, 2017), the present study examines herd behavior during different market episodes of the CSE over a long period. Thus, with the evidence on herd evolution across different market episodes, it provides an explanation for the mix empirical findings on herding, as reported by those previous studies. Fourth, the quantile regression (QR) procedure is employed in this study, in addition to customary used the ordinary least squares (OLS) approach. Hence, the findings contribute the literature in terms of not only robust evidence but also a deeper knowledge of herd behavior in the frontier market setting. Accordingly, the findings will enlarge and enrich the behavioral finance literature, facilitating further understanding of herding phenomenon in financial markets.

> The remainder of the paper is organized as follow. Section 2 reviews the literature about causes of herding and how it would evolve in a financial market. Section 3 explains the nature of the investment environment of the CSE and how it is likely to stimulate herd behavior during the sample period. While discussing about the data in Section 4, Section 5 presents the methodology employed in this study. The empirical results are reported in Section 6. Section 7 concludes the paper with the empirical implications. The limitations of the current study and the direction for future research are given in Section 8.

2. Literature review

Herding is observable as correlated behaviors of a group of investors, occurred consciously or unconsciously, as they trade in the same direction at the same time. (Hwang and Salmon, 2004; Nofsinger and Sias, 1999). It, by becoming a market-wide trend, appears as convergence of investors' behaviors towards consensus at the aggregate market level (Chang et al., 2000). The literature reveals that investors' cognitive limitations, psychological state and external factors such as market conditions, social interactions and information asymmetry are driving forces of herd behavior in financial markets. Some of these causes can be comprehended through the behavioral models of Barberis *et al.* (1998), Daniel et al. (1998), Hong and Stein (1999) and Lam et al. (2010, 2012). These models demonstrate how investor sentiment leads to conservatism and representativeness biases when updating their beliefs. The conservatism bias causes too slow in updating beliefs with new information, thus, overweighs past information while underweighting new information. In case of representativeness bias, investors tend to overweigh the recent information and underweight their past beliefs, thus, update their beliefs too quickly with a small sample of recent observations. Accordingly, the investors who behave too conservatively tend to follow the information implicit in past stock prices, and those who are subject to representativeness bias tend to react based on new information. Hence, their actions can correlate unconsciously due to the similarity of information employed in their decisionmaking. These models also predict that the conservatism and representativeness biases lead to short-term momentum and long-term overreaction and reversal of prices. Then, the momentum traders would enter into market with trend chasing strategies to reap profit opportunities resulting from the short-term price momentum, which can also appear as correlated trading patterns in the market. The resulting excessive momentum in prices and representativeness bias eventually lead to long-term overreactions and subsequent reversal of prices, thereby, increasing market volatility. Consequently, the investors may become unsure of how to react to this increased uncertainty. Thus, they may tend to suppress their own belief and imitate the behaviors of others in the market.

The following sections review the literature relating to causes of herding and how the evolutionary nature of this phenomenon can be expected in financial markets.

2.1 Herding as rational heuristic versus irrational emotional response

Herd behavior occurs as a rational heuristic or an irrational emotional response of market participants. Concerning on the rational side of herding, investment professionals tend to herd intentionally for their reputation or remuneration concerns. Trueman (1994), proposing a model on analyst's forecasting behavior, shows that the earnings forecasts made by analysts do not necessarily contain their private information, and there is a tendency to announce forecasts similar to those made by other analysts to copy higher ability and for higher remuneration. In addition, herding would be occurred unintentionally due to their common education background, use of similar indicators in their analyses or similar stock characteristics employed with investment styles such as value, growth and momentum investing. Further, the theoretical models of Bikhchandani *et al.* (1992) and Banerjee (1992) argue that herding is a rational behavior when other investors have important information relating to one's decisions. Thus, following the others' trading patterns enables to infer such information.

On the other hand, since investors are bounded rational, their sentiment such as optimism, pessimism, fear and ambiguity can affect their investment decision-making. Thus, herding is likely to occur irrationally through such emotions triggered in response to events and conditions in financial markets, for example, information scarcity and asymmetry, extreme volatility or instability, fads and trends (Spyrou, 2013). As many previous studies show[3], investors are optimistic (pessimistic) in rising (falling) markets, hence, are more likely to follow the market trend disregarding the fundamentals. Similar to those of developed and emerging markets, the studies conducted for frontier markets reveal such herd tendencies in both up and down market days. However, the findings are mixed with respect to the asymmetry of herd formation between these up and down market movements. Arjoon and Bhatnagar (2017) on the Trinidad and Tobago Stock Exchange and Rahman et al. (2015) on the Saudi Stock Exchange find a stronger herding in the up-market direction, indicating the fact that investors are more inclined to herd when they are optimistic about market conditions. On the contrary, the findings of Economou (2016) on the stock exchanges of Nigeria and Morocco and Vo and Phan (2016) on the Vietnam stock market reveal that herding is stronger with investors' pessimistic mood during falling market days than their optimistic mood during rising market days. However, Guney et al. (2017) show that herding occurs irrespective of the direction of market movements in the African stock exchanges.

In addition, fear and ambiguity dominate among investors under uncertainty. Consequently, they may react to the uncertainty by herding others' trading patterns. Guney *et al.* (2017) show that herding is strongly present during high volatility days than low volatility days. The study of Balcilar *et al.* (2013) conducted for the Gulf Arab stock markets also suggests that high market volatility influences the formation of herd behavior. Further, the literature reveals that smaller stocks enhance the magnitude of herding due to the greater information uncertainty when trading them (Arjoon and Bhatnagar, 2017; Guney *et al.*, 2017). Hwang and Salmon (2004) and Liao *et al.* (2011) emphasize that emotion is a key driver of herding. Prechter (2001) also shows that sentiment leads investors to follow fads and trends, resulting to herd formation in financial markets. Furthermore, social environment, which is typically characterized by interactions, religious activities, social

Evolution of herd behavior

SEF circles and social conventions, has an influence on herd formation. Hong *et al.* (2004) and Lu and Tang (2015) show how social interactions motivate less sophisticated investors to follow others' strategies, resulting to convergence of investment behavior to a social norm. Gavriilidis *et al.* (2016) find a stronger herd behavior during days of Ramadan than non-Ramadan days, which supports for the fact that social interaction and positive mood are determinants of herding in financial markets.

2.2 Evolution of herding

642

The evolutionary nature of behavioral biases is a research initiative emerged with the AMH of Lo (2004, 2005, 2012). The AMH assumes that market participants are neither perfectly rational nor perfectly irrational. Rather, their rationality is an evolving phenomenon, driven by market forces and their adaptability to market conditions. This evolutionary perspective hypothesizes that they are capable of adapting to market environment by learning about their biases from the reinforcements and the associated emotions experienced with their past behaviors. Thus, the level of rationality depends on the extent to which they are able to learn and select the right behavior for adapting to market conditions. Hence, given enough time and competitive market forces, the market participants with maladaptive behaviors are eliminated from the market population due to losses, and the market will eventually consist of those who adapt to its environment. Consequently, it can expected that behavioral biases decline and disappear over time at the aggregate market level.

In view of that, herd bias could occur with varying magnitude depending on the nature of market environment. In the process of investors' adapting to market conditions, it could evolve to negative herding and absence of herding statuses, as explained below.

2.2.1 Negative herding. The negative herd behavior, also known as "reverse herding" or "anti-herding", takes place when investors act against the market consensus by trading stocks based on views dominant among their subgroups, or at an extreme case, based on each of their own beliefs (Bohl et al., 2017; Gębka and Wohar, 2013). Thus, negative herding symbolizes investors' increased individuality in decision-making (Galariotis et al., 2016). Consequently, stocks are traded excessively based on diverse beliefs of the investors, leading to an increase of price dispersions above the level of rational pricing. Gebka and Wohar (2013) show that the negative herding arises due to phenomena such as investor overconfidence and panic selling induced by uncertain market conditions. In case of overconfidence, following raising market periods, investors tend to attribute high performance of their portfolios to their own skills in stock selection and other abilities rather than to the market conditions. Thus, they are likely to over rely on their own views while ignoring market signals when making subsequent investment decisions. This can result to a high risk-seeking behavior by buying risky stocks and selling low risk investments to greater level. In case of panic selling, during periods of high volatility and uncertainty, irrational fear dominates among investors. Consequently, they become more risk-averse and rebalance their portfolios to more secure positions by selling risky stocks and buying less risk investments to a greater extent. In these instances, stocks would be traded excessively, increasing the price dispersions above the rational level. The former instance leads to a rapid increase of market whereas the latter causes to a substantial market decline.

2.2.2 Absence of herding. This refers to a situation where market participants neither trade by following the market-wide price signals nor exhibit excessive trading as in case of negative herding. It indicates the fact that investor are mostly informed of stock fundamentals. Thus, the trading is based such information, but not merely following the observed actions of other investors or the market-wide price fluctuations. As a result, stocks tend to be traded in a way that results in price dispersions occurring at the rational level.

Certain studies in the literature support the absence of herding status at the aggregate market level. Nguyen (2018) and Yao *et al.* (2014) show that, after periods of significant herding, it declines and becomes absent due to the effectiveness of regulatory reforms in the respective markets. Choe *et al.* (1999) and Hwang and Salmon (2004) find a declining herd tendency following a financial crisis period, which indicates fact that the outbreak of a crisis as a turning point for herd evolution. Arjoon and Bhatnagar (2017) also reveal a strong tendency to herd during periods of high market stress and financial turmoil, which, however, disappears afterwards. They suggest that the regulatory reforms in the aftermath of the crisis to improve the informational environment produce a greater awareness of company fundamentals to investors, hence, lead to the absence of herding in the market.

Alternatively, considering the evolutionary nature of behavioral biases implied by the AMH, the absence of herding can be expected when investors learn the irrationality of herding and pursue the right behavior for adapting to market environment. As the AMH assumes, the learning occurs through the reinforcements and associated emotions experienced from their past behaviors. However, the extant literature casts some doubt on the validity of the reinforcement learning assumption in reducing behavioral biases. If this assumption is valid, a higher level of experience should result to a lower level of behavioral biases. However, previous studies reveal that the experienced investors are more exposed to behavioral biases than the inexperienced investors (Mishra and Metilda (2015), Baker et al. (2019) and references therein). In addition, concerning on the cognitive aging, Korniotis and Kumar (2011) argue that though the older investors have a greater investment knowledge through their experience, their aging impedes the application of their experiences to reduce behavioral biases. Hirshleifer (2015) also claim that reinforcement learning can bias the learning process as investors merely extrapolate their own past experiences, without appropriately reflecting on the experiences when learning, Further, Chiang et al. (2011) and Choi et al. (2009) find the evidence on the investors' tendency to over-extrapolate their past experiences when making decisions.

In view of above evidence, Shantha *et al.* (2018) propose a model of investor learning, which claims that past experiences do not merely result in learning, rather the self-reflection of such experiences facilitates to reduce behavioral biases in decision-making. Thus, the learning occurs when investors cognitively evaluate the validity of mental frames such as beliefs and assumptions underlying their past behaviors by reflecting on the associated experiences. This self-reflection process enables to revise biased mental frames so that future behavior is modified accordingly. Supporting this view, Shantha (2019) find that herd bias is not declined merely by past investment experiences. Rather, investors learn the irrationality of herding through the self-reflection of their experiences so that it is minimized depending on the extent to which they involve in this self-reflection process.

2.3 Previous studies on the Colombo Stock Exchange

There exists a limited number of studies and inconclusive evidence on herd bias in respect of the CSE. According to my best knowledge, there are only three published studies of the market-wide herding on the CSE. It was first examined by Sewwandi (2016) for the entire market and separately for up- and down-market days over the 2001-2015 period, using the Cross Sectional Absolute Deviation methodology suggested by Chang *et al.* (2000) (hereafter called as "CSAD model") with the OLS approach. The results show no evidence of herd behavior during this period. Then, Xiaofang and Shantha (2017) examined it for the 2007-2016 period by applying both the OLS and the QR approaches to the CSAD model. Their OLS findings, consistent with Sewwandi (2016), suggest absence of herding, whereas the results based on the QR procedure provide evidence to support herding in the market during

Evolution of herd behavior

SEF the period. Afterwards, following the same methodology with the OLS approach, Xiaofang and Shantha (2018) reexamined it for the 2000-2016 period. According to their results, herding is strongly evident during the 2000-2008 period, whereas it is absent during the 2008-2016 period. Therefore, studying further about herd behavior on the CSE could provide more insight into these mixed empirical findings reported by those previous studies.

3. The investment environment of the Colombo Stock Exchange and its 644 influence on herding

Currently, the CSE has 298 listed companies, covering 20 business sectors. The total market capitalization as of 31 March 2018 was LKR 3,032.7bn, representing about 22 per cent of the GDP. Similar to most of other frontier markets, the domestic investors presently dominate the CSE, which is about 85 per cent of the total number of shares traded and 96 per cent of the total number of trades executed during the last five years. Further, trading by retail investors represents a significant proportion of the stock trading, which is on average 80 per cent of total number of trades executed.

The CSE, as a pre-emerging market, is regularly exposed to certain risks that cause impediments to its functioning such as regulatory risk, governance risk, liquidity risk, trade execution risk, high vulnerability to economic and environmental shocks and dominance of noise trading. These risks could lead to unfavorable consequences in the market, for example, information asymmetry, insider trading, low transparency, price manipulation, significant mispricing and a few number of securities actively traded, which would discourage investor participation for investing and trading in the market. Consequently, such risks impair both the amount and quality of information incorporated into prices of securities, eventually rendering to a more ambiguous informational environment in the CSE. Accordingly, it is likely that the investors tend to herd for resolving this information uncertainty (Gunev et al., 2017).

The trading environment of the CSE was highly uncertain over the past few decades due to the impacts of economic crises and political instability of the country. The two-decade Civil War worsened the country's economy and political stability. During this period, the market experienced a lower trading volume, which resulted to a low level of liquidity and a slower growth rate in the market. However, since the end of the war in 2009, the market continued with a rapid growth, rising to a speculative bubble which crashed during the period from March 2011 to July 2012. Though the market started to regain its position afterwards, its trading environment has again become highly uncertain since the year 2015 due to the effects of political transitions and economic crisis of the country. Given such stressful market environment, it is also more probable that the investors disregard their own beliefs and are prone to imitate others' behaviors when making their trading decisions (Arjoon and Bhatnagar, 2017; Hwang and Salmon, 2004; Olsen, 2011).

4. Data

36.4

The sample period covers the trading days from April 3, 2000 to March 29, 2018, which consists of a total of 4,305 trading days. A long time window is selected to consider different market episodes of the CSE in the analysis, as shown in Figure 1 and explained below. The time duration from April 2000 to March 2009 represents the period during which the CSE was affected by the political instability of the country as a consequence of the Civil War. Since the end of it, the market showed a rapid growth and was built up to a speculative bubble which crashed during the period from March 2011 to July 2012. The time period subsequent to the crash is split into two sub-periods based on the date of the presidential election, January 8, 2015 as the cut-off point. During the period after the presidential election, the market experienced a slower growth mainly due to the effects of the political instability and the economic crisis of the country. It is expected that such a long time span consisting of different events and conditions affecting the market, provides an appropriate ground for the examination of the evolutionary nature of herding.

The data set includes daily closing prices^[4] of all listed common stocks, which is available in the Data Library (CD ROM) of the CSE. Consistent with the previous studies^[5], both the active and the suspended/delisted stocks during the sample period were included in the sample to alleviate possible survivorship bias affecting the results. A stock's price observations were included in the data set only for the days it was traded so that non-trading or thin trading would have no effects to the results. Accordingly, the final data set includes a total of 321 stocks with 746,518 daily price observations.

5. Methodology

Since this study covers the examination of the evolutionary nature of herding at the market level, a model to detect market-wide herding is employed here. Christie and Huang (1995) and Chang *et al.* (2000) propose respectively the cross-sectional standard deviation (CSSD) model and the CSAD model for detecting the market-wide herd behavior. As the CSSD model examines herding occurred at extreme market movements, it requires defining the extreme lower and upper tails of the return distribution. Consequently, its results are biased towards the definition of the extreme returns and it ignores such behavior occurred during stable market days. Further, it does not account for the nonlinearities of the return in the presence of herding and is sensitive to outliers in data. On the other hand, the CSAD model, as provided by equation (1) below, overcomes these weaknesses[6]. Therefore, it has been the most extensively used model in the market-wide herding studies so far.

$$CSAD_t = \propto +\gamma_1 |R_{mt}| + \gamma_2 R_{mt}^2 + \varepsilon_t \tag{1}$$

In equation (1), R_{mt} and $|R_{mt}|$ stand for the equally weighted cross-sectional average return of all stocks (market portfolio) on day *t* and its absolute value respectively. $CSAD_t$ represents the equally weighted cross-sectional absolute deviation of stock return on day *t*, calculated according to the equation (2).

$$CSAD_t = (1/N) \sum_{i=1}^{N} |R_{it} - R_{mt}|$$
 (2)

In equation (2), R_{it} denotes return of stock *i* on day *t*, determined as the first logarithmic difference of daily closing prices, and *N* is the number of stocks in the market portfolio on day *t*.

5.1 Test of herding and negative herding

When the market participants do not herd, stocks are expected to be traded based on their diverse beliefs about stock fundamentals. Thus, as the Capital Assets Pricing Model implies, the cross sectional absolute deviation of return is a linear and increasing function of the market return. This indicates that the relationship between $CSAD_t$ and R_{mt} is linear and positive, resulting to a significantly positive value for γ_1 and an insignificant value for γ_2 in equation (1).

In the event of herding, since stocks are traded based on information emerged from the market-wide price fluctuations, the stock returns tend to get distributed around the market

Evolution of herd behavior

SEF return, thereby, declining the $CSAD_t$. Accordingly, when herding occurs, the relationship between $CSAD_t$ and R_{mt} becomes negative and non-linear, leading to a significantly negative value for γ_2 in equation (1).

646

Conversely, negative herding occurs when investors trade stocks excessively by acting against the market-wide price fluctuations. It results to an increase of price dispersions above the rational level, which is reflected by a significant positive value of γ_2 in equation (1).

5.2 Test of asymmetric herding effects between up and down market movements

The empirical analysis is further carried out to test whether herding effects are significantly different between the periods of positive market returns (up-market days) and the periods of negative market returns (down-market days). Following the approach proposed by Chiang and Zheng (2010), the equation (3) depicts the extended CSAD model to detect herding asymmetry conditional on the market return.

$$CSAD_{t} = \propto +\gamma_{1}D_{t}^{\mu\rho}|R_{mt}| + \gamma_{2}D_{t}^{\mu\rho}R_{mt}^{2} + \gamma_{3}\left(1 - D_{t}^{\mu\rho}\right)|R_{mt}|$$
$$+ \gamma_{4}\left(1 - D_{t}^{\mu\rho}\right)R_{mt}^{2} + \varepsilon_{t}$$
(3)

In equation (3), D_t^{up} stands for the dummy variable taking the value of 1 for up-market days and zero value for down-market days. Market days are considered to be up- (down-) market when $R_{mt} > 0$ ($R_{mt} < 0$). The Wald test is performed to examine herding asymmetry between these market movements.

5.3 Ordinary least square and quantile regression approaches for estimating the crosssectional absolute deviation model

The regression coefficients of the equations (1) and (3) are customary estimated by applying the OLS approach. The OLS approach requires the parametric assumptions on its error distribution to be satisfied. It is also not an appropriate method when non-linearity exists in the equation and outliers are present in the data[7]. On the other hand, the QR[8] is a semi-parametric regression approach so that it does not necessitate fulfilling the parametric assumptions and is less sensitive to outliers in the data[9]. Hence, the QR is considered to be a more robust method than the OLS regression. In addition, the QR provides a complete picture of how the regression estimates vary over the entire distribution of the dependent variable. Thus, it facilitates to make inferences by examining herding over the entire distribution of $CSAD_t$, especially its magnitude under the extreme market movements. Accordingly, herding estimates are obtained using both the OLS and the QR methods. Equations (4) and (5) represent the QR models for the examination of herding for aggregate market and up-and down-market days respectively[10].

$$Q_{\tau}(\tau | CSAD_t) = \theta_{\tau} + \gamma_{1,\tau} | R_{mt} | + \gamma_{2,\tau} R_{mt}^2$$
(4)

$$Q_{\tau}(\tau | CSAD_{t}) = \theta_{\tau} + \gamma_{1,\tau} D_{t}^{\mu\rho} |R_{mt}| + \gamma_{2,\tau} D_{t}^{\mu\rho} R_{mt}^{2} + \gamma_{3,\tau} \left(1 - D_{t}^{\mu\rho}\right) |R_{mt}|$$

$$+ \gamma_{4,\tau} \left(1 - D_{t}^{\mu\rho}\right) R_{mt}^{2}$$
(5)

where τ represents for quantiles, $\tau \in \{0, \ldots, 1\}$.

The bootstrap technique[11] is employed with the xy-pair design[12] and for 600 replications[13] for computing the confidence intervals of the regression quantiles.

5.4 Hypotheses for examining the evolution of herding

The evolutionary nature of herding is examined by splitting the total sample period into different sub-periods, as shown in Figure 1. Table I shows the results of the Chow's break point test to validate the cut-off points for these sub-periods.

First, the total sample period is equally divided into two sub-periods, consisting of 9 years each. The first nine-year period, covering from April 2000 to March 2009, represents a period during which the country was politically unstable as a result of the Civil War. The high level of uncertainty during this time period considerably affected the flow of investments into and the trading activities of the market, which consequence to a lower liquidity and a slower market growth rate, as reflected by Figure 1. Thus, it can be expected that, in additions to the risks that the CSE is commonly exposed as a frontier market, the market uncertainties caused by the war, in particular, strengthen herd tendency during the 2000-2009 period, as given by the H1.

H1. Investors are prone to herding irrespective of the direction of market movement during the 2000-2009 period so that $\gamma_2 < 0$ in equations (1) and (4), and $\gamma_2 < 0$ and $\gamma_4 < 0$ in equations (3) and (5).

The second 9-year period, spanning from April 2009 to March 2018, is examined by separating into the market bubble period (April 2009 to February 2011), the market crash period (March 2011 to July 2012), and the post-crash period (August 2012 to March 2018). The post-crash period is further split into two periods, namely the periods before and after the presidential election, based on the date of the election (January 8, 2015) as the cut-off point. Accordingly, the second 9-year period consists of four sub-periods, as depicted in Figure 1. The following hypotheses are examined for these sub-periods.

H2. Investors follow the positive price signals emitted in the market since the end of the Civil War in favor of their beliefs about stock fundamentals. Thus, herding would

Sub-periods	Break date	F-statistic	<i>p</i> -value
Periods before and after the Civil War (April 2000 to March 2018) Market bubble and crash periods (April 2009 to July 2012) Periods before and after the presidential election (August 2012 to	1/4/2009 1/3/2011	441.861 18.304	0.000 0.000
March 2018)	8/1/2015	421.911	0.000
Notes: This table reports the results relating to the Chow's break point	at test for ever	mining the w	lidity of

Notes: This table reports the results relating to the Chow's break point test for examining the validity of the cut-off points used in dividing the total sample period into different sub-periods. The significance of *F*-statistic rejects the null hypothesis that there is no break at the specified break point

Table I. Results of the chow's break point test

occur in the up-market direction during the bubble period so that $\gamma_2 < 0$ in equations (1) and (4), and $\gamma_2 < 0$ and $\gamma_4 = 0$ in equations (3) and (5).

- H3. Investors are fear of the uncertainty of the inflated stock prices that occurred during the bubble period. Thus, they act against the market consensus by rebalancing their portfolios to safe positions by selling risky stocks (increased individuality in decision-making). Thus, negative herding would occur in the down-market direction during the market crash period so that $\gamma_2 > 0$ in equations (1) and (4), and $\gamma_2 = 0$ and $\gamma_4 > 0$ in equations (3) and (5).
- H4. Investors learn from the losses occurred due to the irrationality of herding and negative herding. Consequently, they tend to get away from such behavior during the period after the market crash, despite the fact that herd motives such as economic and political uncertainties and high market volatility strongly appear during this post-crash period. Accordingly, $\gamma_2 = 0$ in equations (1) and (4), and $\gamma_2 =$ 0 and $\gamma_4 = 0$ in equations (3) and (5) during the periods before and after the presidential election.

Table II shows a summary of statistics of $CSAD_t$ and R_{mt} for the entire sample period and the sub-periods. Except for the market crash period and the period after the election, the mean value of R_{mt} is consistently positive. It takes the highest value during the bubble period, which reflects strong positive price signals emitting from the market since the end of the war. The standard deviation, which reveals the magnitude of the volatility, of $CSAD_t$ is

Period covered		Mean	SD	Skewness	Jarque–Bera statistic	ADF statistic
Panel A: Total Sa	amble					
	$CSAD_t$	1.036	0.409	2.072	11,208.67***	-6.261^{***}
	R_{mt}	0.053	0.713	0.302	56,297.92***	-45.604^{***}
Panel B: 9-year si	ub-periods					
2000-2009	$CSAD_t$	1.160	0.475	1.858	3,110.50***	-11.204^{***}
	R_{mt}	0.065	0.873	0.231	18,030.52***	-33.088 ***
2009-2018	$CSAD_t$	0.911	0.278	1.328	2,403.99***	-7.238***
	R_{mt}	0.041	0.504	0.368	5,962.32***	-22.797^{***}
Panel C: Analysis	of the 2009)-2018 sub-p	eriod			
Bubble period	$CSAD_t$	0.999	0.248	1.132	195.47***	-4.597 * * *
•	R_{mt}	0.189	0.561	-0.172	251.85***	-18.228^{***}
Crash period	$CSAD_t$	1.080	0.289	1.842	824.79***	-6.298^{***}
1	R _{mt}	-0.059	0.766	0.678	624.59***	-15.496 ***
Before-election	$CSAD_t$	0.981	0.238	0.966	109.74***	-7.670***
	R_{mt}	0.046	0.433	0.255	67.36***	-9.140^{***}
After-election	$CSAD_t$	0.728	0.214	3.156	19751.99***	-6.596^{***}
	R _{mt}	-0.009	0.318	0.120	615.18***	-10.717***

Notes: This table reports statistical properties of the cross-sectional absolute deviation of returns (CSAD_t) and the market return (R_{mt}) series for the sample period from 3/4/2000 to 29/3/2018. Each sub-period given in panel B begins on the April 1 and ends on the March 31 of the given years. Panel C shows the analysis of the 2009-2018 period by split into the market bubble period (April, 2009 to February, 2011), the market crash period (March, 2011 to July, 2012), and the post-crash period (August, 2012 to March, 2018) which is further separated into the periods before and after the presidential election using the date of the election (January 8, 2015) as the cut-off point. The statistical significance at the 1, 5 and 10 per cent levels are Descriptive statistics represented by ***, ** and * respectively

Table II.

SEF

36.4

lower than that of R_{mt} throughout the entire sample period. In addition, the volatility of the series of $CSAD_t$ and R_{mt} are higher during the first half of the sample period than those of the second half. The Augmented Dickey-Fuller (ADF) test statistic rejects the null hypothesis of unit root for both of these variable series, indicating their stationary over the sample period. However, Jarque-Bera statistics show that $CSAD_t$ and R_{mt} series are non-normal. They are skewed to the right, as indicated by the positive skewness statistics. Since the non-normality of the variables may affect the validity of the OLS results, the QR analysis is important here to provide robust evidence to examine the hypotheses.

6. Analysis of results and discussion

The empirical results are reported in the following order. First, the results of the aggregate market are discussed in relation to the total sample period and the sub-periods. Then, those relating to the up and down market days are presented with the evidence on asymmetric herding effects conditional on these market movements. After that, the results are compared over different market periods of the CSE such as period of uncertain market conditions due to the Civil War, market bubble and crash periods and period subsequent to the crash to examine evolution of herding over these market periods. The coefficients, γ_1 in equations (1) and (4), and γ_1 and γ_3 in equations (3) and (5) are significantly positive in both total sample and sub-sample periods' tests, supporting the assumption on the positive relationship between $CSAD_t$ and $|R_{mt}|$.

6.1 Herd behavior at the aggregate market level

Tables III and V present respectively the OLS estimates [estimating equation (1)] and the QR estimates [estimating equation (4)] for examining herd behavior at the aggregate market level. A significantly negative (positive) value of γ_2 indicates herd behavior (negative herding). With respect to the OLS estimates, γ_2 is significantly negative for the total sample

Period covered	x	γ_1	γ_2	Adjusted R ²
Panel A: Total sample				
	0.765 (59.494)***	0.641 (17.189)***	$-0.044(-4.188)^{***}$	0.471
Panel B: 9-year sub-perio	ods			
2000-2009	0.825 (46.552)***	0.651 (13.957)***	-0.046 (-3.828)***	0.481
2009-2018	0.738 (50.260)***	0.536 (11.155)***	-0.048 (-1.590)	0.357
Panel C: Analysis of the	2009-2018 sub-period			
Bubble period	0.839 (33.936)***	0.446 (6.569)***	-0.106(-3.077)***	0.191
Crash period	0.945 (41.341)***	0.205 (4.320)***	0.048 (3.900)***	0.426
Before-election	0.832 (46.582)***	0.494 (6.632)***	-0.070(-1.070)	0.244
After-election	0.568 (40.491)***	0.755 (10.839)***	-0.084(-1.554)	0.504

Notes: This table reports the OLS regression results for the aggregate market, as estimated from equation (1), over the sample period from 3/4/2000 to 29/3/2018. Each sub-period given in panel B begins on the April 1 and ends on the March 31 of the given years. Panel C shows the analysis results of the 2009-2018 period by split into the market bubble period (April, 2009 to February, 2011), the market crash period (March, 2011 to July, 2012), and the post-crash period (August, 2012 to March, 2018) which is further separated into the periods before and after the presidential election based on the date of the election (January 8, 2015) as the cut-off point. The *t*-statistics are computed based on the Newey and West (1987) test for heteroskedasticity and autocorrelation consistent standard errors and given in parentheses. The statistical significance at the 1, 5 and 10 per cent levels are represented by ***, ** and * respectively

Table III. OLS Estimates of herding at the aggregate market level

Evolution of herd behavior

SEF 36,4

650

period and the 2000-2009 sub-period, as shown in panel A and B of Table III. However, it turns into be insignificant during the subsequent period from 2009 to 2018. The QR results are consistent with these OLS estimates. The coefficient γ_2 is significantly negative at all quantile intervals of the total sample period and the 2000-2009 sub-period, as reported in Panel A and B of Table V. However, in relation to the second half of the sample period (2009-2018), γ_2 is significantly negative only at the 0.05 and 0.25 quantile intervals. Thus, according to both the OLS and the QR evidence, herd behavior is strongly evident during the first half of the sample period, which, however, becomes mostly insignificant during the second half. Notably, during the first half, the coefficient γ_2 seems to increase from the median quantile (0.50) to the extreme lower (0.05) and the extreme upper (0.95) quantile intervals, indicating a strong tendency to herd during the extreme market conditions.

These results are not consistent with those found by Sewwandi (2016) and Xiaofang and Shantha (2017) on the CSE. By considering all listed firms, Sewwandi (2016) found no evidence of herding for the period from 2001 to 2015. Subsequently, Xiaofang and Shantha (2017) examined whether investors tend to herd when trading stocks of non-financial firms listed during the period from 2007 to 2016. Their OLS results are consistent with those of the current study for the 2009-2018 period, which reveal no evidence of herd behavior. However, their QR results, contrary to those of the current study, show the presence of herd behavior from the extreme lower to the middle quantile intervals and reverse herding from the middle to the extreme upper quantile intervals[14].

6.2 Herd behavior in up and down market days

Tables IV and VI contain respectively the OLS estimates [estimating equation (3)] and the QR estimates [estimating equation (5)] for examining herd behavior in periods of up and down market movements. Significantly negative (positive) values of the coefficients γ_2 and γ_4 reflect herd behavior (negative herding) in up- and down-market days respectively. As given in Table IV, the OLS estimates indicate herd behavior in both up- and down-market days for the total sample period and the first half of the sample period (2000-2009). Further, it appears to have occurred irrespective of the direction of the market movements since the findings reveal an absence of herd asymmetry between up- and down-market days during this period (as reflected by insignificant Wald test statistics). Nevertheless, the estimates of the second half of the sample period (2009-2018) show herd behavior only for up-market days.

Consistent with the OLS evidence, the QR estimates (Table VI) show herding in both upand down-market days at all quantile intervals of the total sample period (panel A) and the first half period (panel B) with no strong evidence of herd asymmetry between these up- and down-market movements. Further, supporting the QR estimates of the aggregate market, Panel B of Table VI reveals increased herd behavior from the median quantile to the extreme upper (0.95) and the extreme lower (0.05) quantile intervals of both up- and down-market days, indicating a stronger tendency to herd during the extreme market conditions over the 2000-2009 period. However, when the results relating to the second half of the sample period are concerned, consistent with the OLS estimates, herding is mostly absent during downmarket days, as shown in Panel B of Table VI.

Accordingly, both the OLS and the QR results reveal that investors tend to herd strongly in both rising and falling market days over the 2000-2009 period, and get away themselves from such irrational behavior mostly during the falling market days of the subsequent 2009-2018 period. These findings are considerably different from those of the previous studies conducted by Sewwandi (2016) and Xiaofang and Shantha (2017) on the CSE as follows. Applying the OLS procedure, Sewwandi (2016) and Xiaofang and Shantha (2017) found no

Period covered	8	Up-ma γ1	rrket days γ_2	Down-rr Y3	larket days γ_4	Adjusted R^2	Wald test $\gamma_2 - \gamma_4 = 0$
Panel A: Total s.	ample 0.764 (62.840)***	0.685 (21.293)***	-0.043(-5.245)***	0.597 (14.630)***	$-0.049(-5.784)^{***}$	0.478	0.006 [0.343]
Panel B: 9-year : 2000-2009 2009-2018	sub-periods 0.819 (49.878)*** 0.742 (50.916)***	0.696 (17.993) *** 0.619 (12.291) ***	-0.044 (-5.489) *** -0.078 (-2.442) **	$0.632 (12.319)^{***}$ $0.405 (7.962)^{***}$	-0.059(-5.603)*** -0.003(-0.097)	$0.490 \\ 0.371$	0.016[2.036] -0.075 $[10.476]$ ***
Panel C: Analysi Bubble period Crash period Before-election After-election	is of the 2009-2018 s 0.854 (34.619) *** 0.951 (41.042) *** 0.828 (46.096) *** 0.572 (40.346) ***	sub-period 0.522 (7.639)*** 0.278 (5.163)*** 0.523 (7.169)*** 0.873 (11.377)***	$\begin{array}{c} -0.125 \ (-3.213)^{***} \\ 0.018 \ (1.285) \\ -0.064 \ (-1.124) \\ -0.130 \ (-1.400) \end{array}$	0.071 (0.848) 0.124 (2.233)** 0.526 (4.882)*** 0.618 (7.731)***	$\begin{array}{c} 0.021 \ (0.634) \\ 0.081 \ (5.833) *** \\ -0.163 \ (-1.305) \\ -0.033 \ (-0.584) \end{array}$	0.284 0.436 0.247 0.522	-0.146[14.228]*** -0.063[14.249]*** 0.100[0.601] -0.097[1.280]
Notes: This tath 2018. Each sub- period by split. (August, 2012 tr 2015) as the cut- errors and giver and the Chi-sque	ole reports the OLS r period given in pank into the market bub o March, 2018) whici off point. The <i>t</i> -statt a in parentheses. Her are statistics shown i	egression results for el B begins on the A bble period (April, 20 h is further separate istics are computed ding asymmetry bet in square brackets. T	up and down market (pril 1 and ends on the 009 to February, 2011) 2d into the periods befc based on the Newey ar tween up and down ma The statistical significan	lays, as estimated fi March 31 of the giv , the market crash are and after the pre of West (1987) test f rket days is examine to e at the 1, 5 and 10	om equation (3), over t en vears. Panel C show period (March, 2011 to sidential election basec or heteroskedasticity a ed by the Wald test, bar per cent levels are repr	a the sample perior s the analysis July, 2012), an I on the date of and autocorreta sed on the null seented by ***	d from 3/4/2000 to 29/3/ results of the 2009-2018 d the post-crash period the election (January 8, ion consistent standard hypothesis of $\gamma_2 \cdot \gamma_4=0$,*** and * respectively
OLS Estimates of herding conditional on market performance	Table W						Evolution of herd behavior 651

SEF 36,4	Adjusted R^2	0.231 0.235 0.254 0.282 0.379	0.238 0.246 0.273 0.305 0.393	0.223 0.176 0.203 0.195 0.211	2018. Each sub- standard errors
652	γ_2	-0.090 (-3.902)*** -0.044 (-4.383)*** -0.041 (-3.227)*** -0.035 (-2.631)*** -0.088 (-2.847)***	-0.077 (-3.356)*** -0.034 (-3.159)*** -0.041 (-3.062)*** -0.039 (-2.167)**	$\begin{array}{c} -0.186 (.4.481)^{***} \\ -0.154 (.3.814)^{***} \\ -0.065 (.1.239) \\ -0.039 (.1.010) \\ -0.046 (.1.260) \end{array}$	(4), over the sample period from 3/4/2000 to 29/3. The t-statistics are computed based on bootstrap ted by ***, ** and * respectively
	γ_1	0.564 (15.570)*** 0.528 (22.703)*** 0.603 (19.368)*** 0.666 (18.990)*** 1.159 (16.173)***	0.554 (11.426)*** 0.492 (15.261)*** 0.594 (16.842)*** 0.700 (12.204)*** 1.210 (12.2415)***	$0.632 (12.595)^{***}$ $0.618 (11.824)^{***}$ $0.570 (9.526)^{***}$ $0.595 (9.450)^{***}$ $0.603 (7.230)^{***}$	market, as estimated from equation n the 31 st March of the given years. 1%, 5% and 10% levels are represen
	θ_{τ}	0.468 (54.533)*** 0.627 (84.371)**** 0.732 (86.572)**** 0.879 (85.779)**** 1.092 (74.171)***	<pre>>dd; 0.505 (30.678)**** 0.686 (57.077)**** 0.798 (64.575)**** 0.942 (52.218)**** 1.152 (43.262)***</pre>	0.446 (40.990)*** 0.582 (56.099)*** 0.693 (62.781)*** 0.839 (56.110)*** 1.104 (49.244)***	ts the QR results for the aggregate begins on the 1 st April and ends o s. The statistical significance at the
Table V. QR Estimates of herding at the aggregate market level	Quantile	Panel A: Total Sample 0.05 0.25 0.50 0.75 0.95	Panel B: 9-year sub-perid 2000- 2009 sub-period 0.05 0.25 0.50 0.75 0.95	2009- 2018 sub-period 0.05 0.25 0.50 0.75 0.95	Notes: This table repor period given in parel B and given in parenthese

		Up-mai	rket days	Down-n	narket days	1	Wald Test
Quantile	θ_{τ}	γ_I	γ_{2}	γ_{β}	γ_4	Adjusted R^2	$\gamma_2-\gamma_4=0$
<i>Panel A: Tc</i> 0.05 0.25	otal sample period 0.469 (50.649)*** 0.628 (90.063)***	0.678 (14.252)*** 0.607 (14.989)***	-0.134(-4.024)*** -0.055(-2.012)**	0.492 (13.597) *** 0.426 (20.601) ***	-0.067 (-2.974) *** -0.023 (-2.461) ***	0.245 0.249	-0.067 [3.651] -0.032 [2.024]
0.50 0.75 0.95	0.739 (88.942)*** 0.874 (87.572)*** 1.090 (66.668)***	0.651 (19.264) *** 0.734 (17.958) *** 0.734 (17.958) *** 1.160 (10.471) ***	-0.041(-2.278)** -0.042(-2.572)*** -0.088(-1.969)**	0.467 (18.385) *** 0.606 (15.573) *** 1.269 (10.224) *** 1.269 (10.224) ***	-0.025(-2.581)*** -0.044(-4.386)*** -0.150(-2.693)***	$\begin{array}{c} 0.265 \\ 0.287 \\ 0.379 \end{array}$	$\begin{array}{c} -0.016 \left[0.793 \right] \\ 0.002 \left[0.010 \right] \\ 0.063 \left[0.910 \right] \end{array}$
Panel B: 9-, 2000-2009 s 0.05 0.25 0.25 0.25 0.75 0.75 0.75	<i>vab-periods</i> ub-period 0.509 (30.621)*** 0.679 (62.342)*** 0.807 (70.121)*** 0.931 (58.776)**** 1.147 (49.163)***	0.628 (10.487)**** 0.604 (13.151)**** 0.620 (16.808)**** 0.773 (15.848)**** 1.158 (12.684)****	-0.098 (-3.067)*** -0.054 (-2.244)** -0.031 (-1.740)* -0.046 (-2.123)**	0.496 (10.746)*** 0.427 (15.987)*** 0.476 (13.603)*** 0.448 (11.806)*** 1.473 (12.763)***	$\begin{array}{c} -0.063 \left(-2.855\right)^{***} \\ -0.026 \left(-2.710\right)^{***} \\ -0.028 \left(-2.622\right)^{****} \\ -0.028 \left(-2.622\right)^{****} \\ -0.052 \left(-4.215\right)^{****} \\ -0.183 \left(-5.318\right)^{****} \end{array}$	0.250 0.262 0.283 0.312 0.401	-0.035 [1.157] -0.028 [1.064] -0.002 [0.013] 0.006 [0.073] 0.095 [3.315]*
2009-2018 s 0.05 0.25 0.50 0.75 0.95	ub-period 0.443 (49.062)*** 0.583 (56.188)*** 0.702 (60.012)*** 0.850 (58.052)*** 1.110 (48.876)***	0.799 (9.548)*** 0.750 (13.543)*** 0.660 (10.133)*** 0.671 (9.827)*** 0.592 (6.297)***	-0.268 (-3.169)*** -0.206 (-4.509)*** -0.108 (-1.780)* -0.095 (-1.948)* -0.053 (-0.996)	0.566 (10.933) *** 0.484 (10.085) *** 0.405 (5.238) *** 0.370 (5.120) *** 0.545 (4.194) ***	$\begin{array}{c} -0.143 (-3.418)^{***} \\ -0.106 (-2.623)^{***} \\ -0.005 (-0.071) \\ 0.024 (0.425) \\ 0.042 (0.280) \end{array}$	0.236 0.219 0.216 0.206 0.211	-0.125 [2.720]* -0.100 [3.359]* -0.103 [1.631] -0.119 [3.411]* -0.095 [0.375]
Notes: Thi period give given in pa Chi-square s	is table reports the Q n in panel B begins c rentheses. Herding a statistics shown in sc	R results for up and do on the April 1 and ends symmetry between up quare brackets. The sta	wn market days, estimat • on the March 31 of the g and down market days tistical significance at th	ed from equation (5), c given years. The <i>t</i> -stat is examined by the W e 1, 5 and 10 per cent l	ver the sample period fristics are computed base istics are computed base 'ald test, based on the nu evels are represented by	om 3/4/2000 to 29/ ed on bootstrap str ull hypothesis of ***, ** and * resp	3/2018. Each sub- andard errors and $\gamma_2 - \gamma_4 = 0$ and the ectively
herding conditional on market performance	Table VI. QR Estimates of					653	Evolution of herd behavior

evidence of herding in both up and down market days. The QR results of Xiaofang and Shantha (2017) reveal both herding and negative herding during up- and down-market days over the period from 2007 to 2016. In addition, the absence of herd asymmetry between up and down market days during the 2000-2009 period indicates that herding occurs independently from the direction of the market movement (Bekiros *et al.*, 2017; Klein, 2013).

6.3 The evolution of herd behavior over time under different market conditions

SEF

36.4

The evolutionary nature of herd behavior is examined by comparing the evidence over time under different market conditions of the CSE. The empirical results, as discussed in sections 6.1 and 6.2, reveal strong evidence of herd formation irrespective of the direction of the market movement during the first 9 years of the sample period (2000-2009). During this period, the country was politically unstable due to the Civil War, which impacted on trading and investments of the CSE. As Figure 1 exhibits, the market experienced a slower growth during this period since the uncertainties caused by the Civil War led to low trading volume and low liquidity in the market. Consequently, the investors might have disregarded stock fundaments and involved in herding towards more liquid stocks to ensure a higher chance of getting their orders executed in the market. Therefore, the evidence supports the H1, indicating that market uncertainty caused by the political instability induce investors to herd during the 2000-2009 period.

Then, the second half of the sample period (2009-2018) characterizes uncertain market episodes such as the market bubble, subsequent crash and the post-crash period affected by political and economic crises of the country. Though the literature suggests that uncertainty and high market volatility are motives for either herding or negative herding, the estimates given in panel B of Tables III, IV, V, and VI provide no strong support to this prediction. During the 2009-2018 period, while there is no evidence of negative herding, herding appears to be more towards the up-market direction, however, less intensely when compared to the 2000-2009 period.

A better picture of herd behavior during the 2009-2018 period can be gained when the estimates obtained separately for the market bubble, the crash and the post-crash subperiods, as shown in Tables III, IV and VII, are considered in the analysis. They indicate changes in the directionality of herding between the bubble and the crash periods. Both the OLS and the QR estimates at the aggregate market level, as given in panel C of Tables III and VII, reveal strong evidence of herd formation during the bubble period (as indicated by significantly negative γ_2), whereas negative herding during the crash period (as indicated by significantly positive γ_2). In addition, panel C of Table IV shows that herd formation occurs during up-market days of the bubble period, whereas negative herding appears during down-market days of the crash period. Accordingly, the evidence supports the H_2 , indicating the investors' tendency to ignore their own views and trade stocks in favor of the positive price signals emanated in the market from the end of the Civil War, which resulted to a speculative bubble in the market (Bekiros *et al.*, 2017). Further, the evidence on negative herding during the crash period supports the H3, revealing the increased individuality in decision-making of the investors. Thus, during the crash period, investors would have relied more on their individual views about the uncertainty of inflated stock prices during the bubble period and resort to panic selling for rebalancing their portfolios to more secure positions (Gebka and Wohar, 2013). This panic selling caused to a declining market, eventually leading to the market crash.

Herding during the bubble period, as discussed above, can be attributed to the effects of investors' conservatism and representativeness biases and momentum trading. Lam *et al.* (2010, 2012) demonstrate that investors are affected by conservatism and representativeness

Quantile (%)	heta au	γ_1	γ_2	Adjusted R^2	Evolution of
Panel A: Market	t Bubble period (April 1	2009 to Februarv 28. 20	011)		HEI U DEHAVIOI
5	0.596 (26.029)***	0.314 (4.063)***	-0.068(-1.556)	0.112	
25	0.713 (52.459)***	0.348 (8.002)***	-0.074 (-3.071)***	0.115	
50	0.804 (36.603)***	0.486 (6.333)***	-0.138 (-3.233)***	0.100	
75	0.937 (29.175)***	0.534 (5.262)***	-0.136 (-2.215)**	0.098	
95	1.206 (16.409)***	0.457 (1.878)*	-0.068 (-0.352)	0.108	655
Panel B: Market	Crash period (March 1,	2011 to July 31, 2012)			
5	0.696 (28.161)***	0.076 (1.578)	0.087 (4.312)***	0.129	
25	0.834 (32.243)***	0.090 (1.556)	0.076 (3.110)***	0.138	
50	0.935 (39.551)***	0.189 (3.120)***	0.048 (2.080)**	0.171	
75	1.069 (37.409)***	0.216 (2.525)**	0.048 (1.638)	0.219	
95	1.207 (21.103)***	0.433 (2.734)***	-0.011 (-0.217)	0.364	
Panel C: Post-cro	ash Period - before the ele	ection (August 1, 2012 t	o January 8, 2015)		
5	0.608 (30.508)***	0.383 (3.965)***	-0.093 (-1.046)	0.114	
25	0.698 (54.066)***	0.483 (6.088)***	-0.087(-0.962)	0.142	
50	0.793 (42.598)***	0.456 (4.824)***	-0.051(-0.541)	0.144	
75	0.902 (34.076)***	0.676 (5.666)***	-0.110(-0.981)	0.139	
95	1.204 (20.602)***	0.645 (2.920)***	-0.165 (-0.873)	0.120	
Panel D: Post-cr	ash Period- after the elec	tion (January 9, 2015 to	March 31, 2018)		
5	0.393 (33.538)***	0.719 (8.302)***	-0.186(-1.643)	0.244	
25	0.488 (50.188)***	0.723 (10.755)***	-0.136 (-1.916)*	0.264	
50	0.572 (59.983)***	0.720 (11.434)***	-0.109(-1.500)	0.282	
75	0.648 (70.177)***	0.715 (9.248)***	-0.067(-0.728)	0.329	
95	0.806 (16.882)***	0.347 (0.728)	0.718 (0.708)	0.394	Table VII.
					QR Estimates of
Notea: This tal	ble reports the QR result	s, as estimated from eq	uation (4), for the market b	ubble period, the	herding for the
market crash pe	eriod and the post-crash	period which is furthe	r split into the periods bef	ore and after the	

Notea: This table reports the QR results, as estimated from equation (4), for the market bubble period, the market crash period and the post-crash period which is further split into the periods before and after the presidential election based on the date of the election (January 8, 2015) as the cut-off point. The *t*-statistics are computed based on bootstrap standard errors and given in parentheses. The statistical significance at the 1, 5 and 10 per cent levels are represented by ***, ** and * respectively

QR Estimates of herding for the market bubble, the crash and the postcrash periods

biases at the same time. Accordingly, after the Civil War, investors are optimistic about the country's economy and political stability. However, their conservatism leads to slow updating of their beliefs, thus, tends to underreact to this structural change. As a consequence, momentum trading builds up in the market to profit from this under-reaction, resulting to excessive trading in the direction of this new information. Their representativeness bias causes for continuing this herd formation over the long-term, which leads to overreaction of prices to that structural change, forming a speculative bubble in the CSE. Then, subsequent price reversals result to burst of the bubble. Guo *et al.* (2017) show that such short-term momentum, long-term overreaction and reversal anomalies persist during periods of financial crisis. In view of that, herd behavior appears to be a reason for the crisis period of the CSE.

Notably, the evidence indicates the absence of herding subsequent to the market crash. The OLS estimates given in Panel C of Tables III and IV and the QR estimates given in Panel C and D of Table VII consistently show that both herding and negative herding are completely absent after the market crash (during both before and after the presidential election) even though herd motives such as market uncertainties caused by economic and political instability of the country strongly appear during this period. The previous

SEF 36,4 studies[15] suggest regulatory reforms in the aftermath of a crisis as a reason for the absence of herding. However, there were no significant regulatory reforms in the CSE during this post-crash period. Hence, the absence of herding could be attributed to the evolutionary nature of behavioral biases, as predicted by the AMH. It implies that investors may have learned from their losses due to their irrational herding and negative herding, and adapted to a more rational behavior over time. The study conducted by Shantha (2019) on the CSE shows the evidence in support of this learning behavior at the individual investor level. Therefore, with the consideration of these empirical results, there is a strong evidence to confirm the H4.

7. Conclusion and implications

The aim of this study is to examine the evolution of herd behavior in a frontier stock market, the CSE of Sri Lanka. The findings, based on both the OLS and the QR procedures, reveal a strong herd tendency at the aggregate market level and up and down market days during the first half of the sample period (2000-2009). However, there is no evidence to support herd asymmetry between up and down market days. In addition, the magnitude of herding appears to increase during the days of the extreme market movements over this period. Accordingly, it is evident that the investors involved in herding irrespective of the direction of the market movement, however, more strongly at the extreme market conditions in the 2000-2009 period. During this first-half period, the market was highly uncertain as a consequence of the effect of the political instability of the country due to the Civil War. Hence, the results support for the general prediction that market uncertainties induce investors to imitate others' behaviors, while disregarding their own beliefs and information when trading stocks.

Contrary to the first half of the sample period, the results of the second half (2009-2018) reveal significant changes to herding pattern over different market episodes. On one hand, herding is evident during up-market days of the bubble period, whereas negative herding during down-market days when the bubble's burst occurred in the market. Thus, the results support the findings of the previous studies which indicate that herding causes unfavorable effects to financial markets such as bubbles and crashes (Spyrou, 2013). On the other hand, either herding or negative herding is not consistently evident during the period after the market crash, despite its motives, especially, market uncertainties resulting from the political instability and the economic crisis of the country occurred during that period. Accordingly, the findings support for the evolutionary nature of behavioral biases, as predicted by the AMH. Based on this evolutionary perspective and the findings of Shantha (2019), the absence of herding during the period after the market crash implies the investors' learning about the irrationality of herding and negative herding from their financial losses and associated psychological shocks such as panic and frustration experienced during the previous market periods. Consequently, they inclined to get away from such irrational behaviors subsequently.

Therefore, consistent with Bekiros *et al.* (2017), Klein (2013), Zhou and Anderson (2013) and Clements *et al.* (2017), herding is a dynamic phenomenon in a financial market. Hence, it is likely to evolve over time with period-specific characteristics of a market. As the AMH assumes, it could decline and disappear over time through the competition and adaptation of market participants' behaviors to market environment. Further, the evidence on the evolutionary nature of herding can be used as an explanation for the mix empirical findings on herding, as reported by the previous studies on the CSE (Sewwandi, 2016; Xiaofang and Shantha, 2017, 2018).

8. Limitations and future research directions

This paper provides insights on the evolutionary nature of herding at the aggregate market level, as a novel attempt for contributing the herding literature. Despite the selection of an appropriate market setting with a suitable study approach, the contribution of this paper is limited in respect of the following. First, the current literature demonstrates a set of market characteristics as determinants of herding, for example, stock size, trading volume, volatility, industry types and market integration (Spyrou, 2013). Herd evolution could also be a function of these factors. For instance, it would be significantly different between small vs large stocks, high trading volume days vs low trading volume days, high volatility days vs low volatility days and among different industry types. However, this study does not cover the examination of the evolution of herding conditional on such market characteristics. Second, the current study examines herd evolution at the aggregate market level. However, it could vary among different investor-types (for example, retail vs institutional investors) and different investment styles (for example, value vs. growth investing), and be related to factors other than investors' learning (for example, changes in trading rules and changes in investor composition). Accordingly, future studies could contribute the literature by examining the nature of herd evolution conditional on such micro and macro characteristics of a financial market.

Notes

- 1. The frontier equity indices provided by Morgan Stanley Capital International (MSCI), Russell, Standard & Poor and FTSE consistently classify the CSE as a frontier market.
- 2. See, for example, Arjoon and Bhatnagar (2017) and Yao, et al., (2014).
- 3. See the literature reviews of Bikhchandani and Sharma (2000), Hirshleifer and Hong (2003) and Spyrou (2013).
- 4. The daily prices are used to test herd behavior since the findings of Christie and Huang (1995) and Tan, Chiang, Mason, and Nelling (2008) suggest that herding is more evident with the daily frequencies of data than weekly or monthly frequencies.
- 5. See, for example, Gavriilidis et al. (2016) and Guney et al. (2017).
- See discussions of Bekiros, Jlassi, Lucey, Naoui, and Uddin (2017), T. C. Chiang *et al.*, (2010), Economou *et al.*, (2011), Focardi *et al.*, (2002) and Lux (1995) for weaknesses of the CSSD model and how they are overcome with the CSAD model.
- 7. See, for example, Barnes and Hughes (2002), Wilcox (2010) and Zhou and Anderson (2013) for statistical assumptions and weaknesses of the OLS regression.
- Since the introduction of the QR by Koenker and Bassett (1978), it has been widely applied for econometrics and financial researches(Allen *et al.*, 2012). However, its application for the analysis of herding is very limited (Bekiros *et al.*, 2017).
- 9. See the discussion on the QR provided by Alexander (2008), Allen *et al.* (2012), Barnes and Hughes (2002), Cade and Noon (2003) and Xiao (2012).
- 10. The Eviews® Version 8 (2013) is used for obtaining the QR estimates.
- 11. The studies conducted by Chen and Wei (2005); Kocherginsky, He and Mu (2005); Koenker (1994); Koenker and Hallock (2000) provide a comparison of different methods available for determining confidence intervals under the QR procedure. They find that these methods slightly deviate from each other, however suggest the bootstrap technique as appropriate one for a study employing sample of less than 5,000 observations and less than 20 covariates.

Evolution of herd behavior

36,4	12.	Buchinsky (1995) and Machado and Silva (2013) find that the xy-pair bootstrap estimator enables for computing more valid standard errors than the other bootstrap estimators.
	13.	This procedure is consistent with Andrews and Buchinsky (2000) and Koenker and Hallock (2000).
658	14.	The current study is not comparable to Xiaofang and Shantha (2017) in terms of the study sample since the former covers all the stocks whereas the latter considers only the stocks of the non-financial companies listed on the CSE. Thus, the contradictory evidence between the two studies may be due to the differences in the study samples.
	15.	See, for example, Arjoon and Bhatnagar (2017), Nguyen (2018) and Yao et al. (2014).

References

CEE

- Alexander, C. (2008), "Market risk analysis", *Quantitative Methods in Finance*, West Sussex, Vol. 1, p. 23.
- Allen, D.E., Singh, A.K., Powell, R.J., McAleer, M., Taylor, J. and Thomas, L. (2012), "The volatilityreturn relationship: insights from linear and non-linear quantile regressions", Working Paper No. 1201, Edith Cowan University, Australia.
- Andrews, D.W. and Buchinsky, M. (2000), "A three-step method for choosing the number of bootstrap repetitions", *Econometrica*, Vol. 68 No. 1, pp. 23-51.
- Arjoon, V. and Bhatnagar, C.S. (2017), "Dynamic herding analysis in a frontier market", *Research in International Business and Finance*, Vol. 42, pp. 496-508.
- Baker, H.K., Kumar, S., Goyal, N. and Gaur, V. (2019), "How financial literacy and demographic variables relate to behavioral biases", *Managerial Finance*, Vol. 45 No. 1, pp. 124-146.
- Balcilar, M., Demirer, R. and Hammoudeh, S. (2013), "Investor herds and regime-switching: evidence from Gulf Arab stock markets", *Journal of International Financial Markets, Institutions and Money*, Vol. 23, pp. 295-321.
- Banerjee, A.V. (1992), "A simple model of herd behavior", The Quarterly Journal of Economics, Vol. 107 No. 3, pp. 797-817.
- Barberis, N., Shleifer, A. and Vishny, R. (1998), "A model of investor sentiment", *Journal of Financial Economics*, Vol. 49 No. 3, pp. 307-343.
- Barnes, M.L. and Hughes, A.T.W. (2002), "A quantile regression analysis of the cross section of stock market returns", Working Paper series No. 02-2, Federal Reserve Bank of Boston.
- Bekiros, S., Jlassi, M., Lucey, B., Naoui, K. and Uddin, G.S. (2017), "Herding behavior, market sentiment and volatility: will the bubble resume?", *The North American Journal of Economics and Finance*, Vol. 42, pp. 107-131.
- Bikhchandani, S. and Sharma, S. (2000), "Herd behavior in financial markets", IMF Working Papers, Vol. 47 No. 3, pp. 279-310.
- Bikhchandani, S., Hirshleifer, D. and Welch, I. (1992), "A theory of fads, fashion, custom, and cultural change as informational Cascades", *Journal of Political Economy*, Vol. 100 No. 5, pp. 992-1026.
- Bohl, M.T., Branger, N. and Trede, M. (2017), "The case for herding is stronger than you think", *Journal* of Banking and Finance, Vol. 85, pp. 30-40.
- Buchinsky, M. (1995), "Estimating the asymptotic covariance matrix for quantile regression models a monte Carlo study", *Journal of Econometrics*, Vol. 68 No. 2, pp. 303-338.
- Cade, B.S. and Noon, B.R. (2003), "A gentle introduction to quantile regression for ecologists", Frontiers in Ecology and the Environment, Vol. 1 No. 8, pp. 412-420.
- Chang, E.C., Cheng, J.W. and Khorana, A. (2000), "An examination of herd behavior in equity markets: an international perspective", *Journal of Banking and Finance*, Vol. 24 No. 10, pp. 1651-1679.

Chen, C. and Wei, Y. (2005), "Computational issues for quantile regression", Sankhyā: The Indian Journal of Statistics, Vol. 67 No. 2, pp. 399-417.

- Chiang, T.C. and Zheng, D. (2010), "An empirical analysis of herd behavior in global stock markets", *Journal of Banking and Finance*, Vol. 34 No. 8, pp. 1911-1921.
- Chiang, T.C., Li, J. and Tan, L. (2010), "Empirical investigation of herding behavior in Chinese stock markets: evidence from quantile regression analysis", *Global Finance Journal*, Vol. 21 No. 1, pp. 111-124.
- Chiang, Y.M., Hirshleifer, D., Qian, Y. and Sherman, A.E. (2011), "Do investors learn from experience? Evidence from frequent IPO investors", *Review of Financial Studies*, Vol. 24 No. 5, pp. 1560-1589.
- Choe, H., Kho, B.-C. and Stulz, R.M. (1999), "Do foreign investors destabilize stock markets? The Korean experience in 1997", *Journal of Financial Economics*, Vol. 54 No. 2, pp. 227-264.
- Choi, J.J., Laibson, D. and Madrian, B.C. (2009), "Mental accounting in portfolio choice: evidence from a flypaper effect", *American Economic Review*, Vol. 99 No. 5, pp. 2085-2095.
- Christie, W.G. and Huang, R.D. (1995), "Following the pied piper: do individual returns herd around the market?", *Financial Analysts Journal*, Vol. 51 No. 4, pp. 31-37.
- Clements, A., Hurn, S. and Shi, S. (2017), "An empirical investigation of herding in the US stock market", *Economic Modelling*, Vol. 67, pp. 184-192.
- Daniel, K., Hirshleifer, D. and Subrahmanyam, A. (1998), "Investor psychology and security market under-and overreactions", *The Journal of Finance*, Vol. 53 No. 6, pp. 1839-1885.
- Economou, F. (2016), "Herd behavior in frontier markets: evidence from Nigeria and Morocco", *Handbook of Frontier Markets*, Elsevier, pp. 55-69.
- Economou, F., Kostakis, A. and Philippas, N. (2011), "Cross-country effects in herding behaviour: evidence from four South European markets", *Journal of International Financial Markets*, *Institutions and Money*, Vol. 21 No. 3, pp. 443-460.
- Focardi, S., Cincotti, S. and Marchesi, M. (2002), "Self-organization and market crashes", Journal of Economic Behavior and Organization, Vol. 49 No. 2, pp. 241-267.
- Galariotis, E.C., Krokida, S.-I. and Spyrou, S.I. (2016), "Herd behavior and equity market liquidity: evidence from major markets", *International Review of Financial Analysis*, Vol. 48, pp. 140-149.
- Gavriilidis, K., Kallinterakis, V. and Tsalavoutas, I. (2016), "Investor mood, herding and the Ramadan effect", *Journal of Economic Behavior and Organization*, Vol. 132, pp. 23-38.
- Gębka, B. and Wohar, M.E. (2013), "International herding: Does it differ across sectors?", Journal of International Financial Markets, Institutions and Money, Vol. 23, pp. 55-84, doi: 10.1016/j. intfin.2012.09.003.
- Guney, Y., Kallinterakis, V. and Komba, G. (2017), "Herding in frontier markets: evidence from African stock exchanges", *Journal of International Financial Markets, Institutions and Money*, Vol. 47, pp. 152-175.
- Guo, X., McAleer, M., Wong, W.-K. and Zhu, L. (2017), "A Bayesian approach to excess volatility, shortterm underreaction and long-term overreaction during financial crises", *The North American Journal of Economics and Finance*, Vol. 42, pp. 346-358.
- Hirshleifer, D. (2015), "Behavioral finance", Annual Review of Financial Economics, Vol. 7 No. 1, pp. 133-159.
- Hirshleifer, D. and Hong, T.S. (2003), "Herd behaviour and cascading in capital markets: a review and synthesis", *European Financial Management*, Vol. 9 No. 1, pp. 25-66.
- Hong, H. and Stein, J.C. (1999), "A unified theory of underreaction, momentum trading, and overreaction in asset markets", *The Journal of Finance*, Vol. 54 No. 6, pp. 2143-2184.
- Hong, H., Kubik, J.D. and Stein, J.C. (2004), "Social interaction and stock-market participation", *The Journal of Finance*, Vol. 59 No. 1, pp. 137-163.

Evolution of herd behavior

SEF	Hwang, S. and Salmon, M. (2004), "Market stress and herding", Journal of Empirical Finance, Vol. 11 No. 4, pp. 585-616.
50,4	Klein, A.C. (2013), "Time-variations in herding behavior: evidence from a Markov switching SUR model", Journal of International Financial Markets, Institutions and Money, Vol. 26, pp. 291-304.
660	Kocherginsky, M., He, X. and Mu, Y. (2005), "Practical confidence intervals for regression quantiles", <i>Journal of Computational and Graphical Statistics</i> , Vol. 14 No. 1, pp. 41-55.
	Koenker, R. (1994), "Confidence intervals for regression quantiles. Paper presented at the 5th prague symposium on asymptotic statistics", Physica, Springer, Heidelberg, pp. 349-359.
	Koenker, R. and Bassett, J.G. (1978), "Regression quantiles", Econometrica, Vol. 46 No. 1, pp. 33-50.
	Koenker, R. and Hallock, K. (2000), "Quantile regression: an introduction", Journal of Economic Perspectives, Vol. 15 No. 4, pp. 43-56.
	Korniotis, G.M. and Kumar, A. (2011), "Do older investors make better investment decisions?" The Review of Economics and Statistics, Vol. 93 No. 1, pp. 244-265.
	Lam, K., Liu, T. and Wong, WK. (2010), "A pseudo-Bayesian model in financial decision making with implications to market volatility, under-and overreaction", <i>European Journal of Operational Research</i> , Vol. 203 No. 1, pp. 166-175.
	Lam, K., Liu, T. and Wong, WK. (2012), "A new pseudo-Bayesian model with implications for financial anomalies and investors' behavior", <i>Journal of Behavioral Finance</i> , Vol. 13 No. 2, pp. 93-107.
	Liao, TL., Huang, CJ. and Wu, CY. (2011), "Do fund managers herd to counter investor sentiment?", Journal of Business Research, Vol. 64 No. 2, pp. 207-212.
	Lo, A.W. (2004), "The adaptive markets hypothesis: market efficiency from an evolutionary perspective", <i>The Journal of Portfolio Management</i> , Vol. 30 No. 5, pp. 15-29.
	Lo, A.W. (2005), "Reconciling efficient markets with behavioral finance: the adaptive markets hypothesis", <i>Journal of Investment Consulting</i> , Vol. 7 No. 2, pp. 21-44.
	Lo, A.W. (2012), "Adaptive markets and the new world order", <i>Financial Analysts Journal</i> , Vol. 68 No. 2, pp. 18-29.
	Lu, T.J. and Tang, N. (2015), "Social interaction effects and individual portfolio choice: evidence from 401 (k) pension plan investors", Wharton Pension Research Council Working Papers, p. 69.
	Lux, T. (1995), "Herd behaviour, bubbles and crashes", <i>The Economic Journal</i> , Vol. 105 No. 431, pp. 881-896.
	Machado, J.A. and Silva, J. (2013), Quantile Regression and Heteroskedasticity, Unpublished manuscrip, Department of Economics, University of Essex, available at; https://jmcss.som.surrey.ac.uk/ JM_JSS.pdf
	Mishra, K. and Metilda, M.J. (2015), "A study on the impact of investment experience, gender, and level of education on overconfidence and self-attribution bias", <i>IIMB Management Review</i> , Vol. 27 No. 4, pp. 228-239.
	Newey, W.K. and West, K.D. (1987), "A simple, positive semi-definite, heteroskedasticity and autocorrelation consistent covariance matrix", <i>Econometrica</i> , Vol. 55 No. 3, pp. 703-708.
	Nguyen, T.T. (2018), <i>Herding Behaviour in Vietnamese Stock Market</i> , Hanken School of Economics. Department of Finance and Statistics, available at: https://helda.helsinki.fi/dhanken/bitstream/ handle/123456789/202979/nguyen.pdf?sequence=1
	Nofsinger, J.R. and Sias, R.W. (1999), "Herding and feedback trading by institutional and individual investors", <i>The Journal of Finance</i> , Vol. 54 No. 6, pp. 2263-2295.
	Olsen, R.A. (2011), "Financial risk perceptions: a behavioral perspective advances", in <i>Entrepreneurial Finance</i> , Springer, New York, NY, pp. 45-67.

Prechter, R.R. (2001), "Unconscious herding behavior as the psychological basis of financial market trends and patterns", *Journal of Psychology and Financial Markets*, Vol. 2 No. 3, pp. 120-125.

661

- Rahman, M.A., Chowdhury, S.S.H. and Sadique, M.S. (2015), "Herding where retail investors dominate trading: the case of Saudi Arabia", *The Quarterly Review of Economics and Finance*, Vol. 57, pp. 46-60.
- Sewwandi, W.G.T. (2016), "Herding in Colombo stock exchange", EPRA International Journal of Multidisciplinary Research, Vol. 2 No. 3, pp. 182-186.
- Shantha, K.V.A. (2019), "Individual investors' learning behavior and its impact on their herd bias: an integrated analysis in the context of stock trading", *Sustainability*, Vol. 11 No. 5, pp. 1448.
- Shantha, K.V.A., Xiaofang, C. and Gamini, L.P.S. (2018), "A conceptual framework on individual investors' learning behavior in the context of stock trading: an integrated perspective", *Cogent Economics and Finance*, Vol. 6 No. 1, pp. 1-22, doi: 10.1080/23322039.2018.1544062.
- Spyrou, S. (2013), "Herding in financial markets: a review of the literature", *Review of Behavioural Finance*, Vol. 5 No. 2, pp. 175-194.
- Tan, L., Chiang, T.C., Mason, J.R. and Nelling, E. (2008), "Herding behavior in Chinese stock markets: an examination of a and B shares", *Pacific-Basin Finance Journal*, Vol. 16 Nos 1/2, pp. 61-77.
- Trueman, B. (1994), "Analyst forecasts and herding behavior", *Review of Financial Studies*, Vol. 7 No. 1, pp. 97-124.
- Vo, X.V. and Phan, D.B.A. (2016), "Herd behavior in emerging equity markets: evidence from vietnam", Asian Journal of Law and Economics, Vol. 7 No. 3, pp. 369-383.
- Wilcox, R.R. (2010), Fundamentals of Modern Statistical Methods: Substantially Improving Power and Accuracy, Springer Science & Business Media.
- Xiao, Z. (2012), "Time series quantile regressions", in Rao, S.R.T. and Rao, C. (Eds), Handbook of Statistics, Time Series Analysis: Methods and Applications, Vol. 30, Elsevier, Amsterdam, pp. 213-257.
- Xiaofang, C. and Shantha, K.V.A. (2017), "Empirical study on herding behavior in Sri Lankan stock market: evidence from quantile regression approach", Paper presented at the 14th International Conference on Innovation and Management, Wuhan University of Technology Press, Wuhan, China.
- Xiaofang, C. and Shantha, K.V.A. (2018), "Are market performance and volatility determining the evolution of herd mentality of investors in a frontier stock market? Evidence from the Colombo stock exchange of Sri Lanka", *Journal of Management Matters*, Vol. 5 No. 1, pp. 1-12.
- Yao, J., Ma, C. and He, W.P. (2014), "Investor herding behaviour of Chinese stock market", International Review of Economics and Finance, Vol. 29, pp. 12-29.
- Zhou, J. and Anderson, R.I. (2013), "An empirical investigation of herding behavior in the US REIT market", *The Journal of Real Estate Finance and Economics*, Vol. 47 No. 1, pp. 83-108.

Corresponding author

Kalugala Vidanalage Aruna Shantha can be contacted at: shantharuna@yahoo.com

For instructions on how to order reprints of this article, please visit our website: **www.emeraldgrouppublishing.com/licensing/reprints.htm** Or contact us for further details: **permissions@emeraldinsight.com**